Author's response to reviews

Title: Self-reported racial discrimination, response to unfair treatment, and coronary calcification in asymptomatic adults - the North Texas Healthy Heart Study

Authors:

Roberto Cardarelli (rcardare@hsc.unt.edu)
Kathryn M Cardarelli (kcardare@hsc.unt.edu)
Kimberly G Fulda (kfulda@hsc.unt.edu)
Anna Espinoza (aespinoz@hsc.unt.edu)
Clifton Cage (ccage@hsc.unt.edu)
Jamboor Vishwanatha (jvishwan@hsc.unt.edu)
Richard Young (ryoung01@jpshealth.org)
Darryl N Steele (darrylsteelemd@yahoo.com)
Joan Carroll (jcarroll@hsc.unt.edu)

Version: 3 Date: 7 May 2010

Author's response to reviews:

All revision were made per the recommendation of the BMC Editorial team.